Author's response to reviews

Title: Health system outcomes and determinants amenable to Public Health in industrialized countries: a pooled, cross-sectional time series analysis

Authors:

Onyebuchi A. Arah (o.a.arah@amc.uva.nl)
Gert P. Westert (gert.westert@rivm.nl)
Diana M. J. Delnoij (d.delnoij@nivel.nl)
Niek S. Klazinga (n.s.klazinga@amc.uva.nl)

Version: 3 Date: 19 July 2005

Author's response to reviews: see over
Dear Editorial Team,

**Health System Outcomes and Determinants Amenable to Public Health in Industrialized Countries: A Pooled, Cross-Sectional Time Series Analysis**

We are delighted by your willingness to accept our manuscript, “Health System Outcomes and Determinants Amenable to Public Health in Industrialized Countries: A Pooled, Cross-Sectional Time Series Analysis” for publication as a Research Article in your esteemed journal. We confirm that the paper has not been previously published, nor is it being considered by any other journal.

We have made following minor changes according to your directions:
1. matched authors’ names and order with the submission system
2. formatted section headings and referencing
3. removed the website (www.fao.org) from the manuscript text and placed it in the reference list thus increasing the number of references from the previous 63 to 64 now
4. removed the text ‘place Table here’ from the manuscript
5. included solid black borders for the Tables
6. added the suggested sentence for our lack of conflicting interests

We look forward to a final, favorable decision to publish this interesting manuscript. We will gladly assist in any way you deem necessary. Dr. Arah will handle all correspondence about this paper.

Thank you for your professional support.

Sincerely yours,

Onyebuchi A. Arah, MPH, DSc, MB BS

Gert P. Westert, PhD

Diana M. Delnoij, PhD

Niek S. Klazinga, PhD, MD